Author’s response to reviews

Title: A comparison of zero-profile anchored spacer (ROI-C) and plate fixation in 2-level noncontiguous anterior cervical discectomy and fusion- a retrospective study

Authors:

Zongyu Zhang (414316457@qq.com)

Yawei Li (20105232050@suda.edu.cn)

Weimin Jiang (jwmpaper@126.com)

Version: 4 Date: 28 Dec 2017

Author’s response to reviews:

Dear editors:

Thank you for your letter and comments concerning our manuscript entitled “A comparison of zero-profile anchored spacer (ROI-C) and plate fixation in 2-level noncontiguous anterior cervical discectomy and fusion- a retrospective study”. The comments are very valuable for improving our paper. We have studied comments carefully and made correction which we hope meet with approval. The corrections in the paper and the responds to your comments are as following:

Responds to the Editor’s comments:

Editor Comments:

I would like to ask the authors, however, to change the group names into group Plate and group NoPlate (instead of B and A) as this would facilitate reading a lot.

Response:

Thank you for your suggestion and we have made the corrections. (line 21,23,33,34,80,81,94,96,149-152,158,159,163,164,169,170,174,175,178,184,185,217,218,231,232,355 and Table 1,3,4)
Thank you very much for your comments and suggestions!

Yours sincerely,

Weimin Jiang